Author's response to reviews

**Title:** Differentials of fertility in North and South Gondar zones, Northwest Ethiopia: A comparative cross-sectional study

**Authors:**

Getu D. Alene (adgetu123@yahoo.com)
Alemayehu Worku (alemayehuwy@yahoo.com)

**Version:** 7 **Date:** 28 November 2008

**Author's response to reviews:**

Reference 12 is cited in text as requested by The Biomedical Editorial Production Team.